Author's response to reviews

Title: Mechanism underlying antidiarrhoeal, antispasmodic and bronchodilator activities of Ipomoea turpethum Linn.

Authors:

    Huma Shareef (phr_huma@hotmail.com)
    Ghazala H Rizwani (rizwanix2@yahoo.com)
    Safur Rehman Mandukhail (saifmandukhal@yahoo.com)
    Naoharu Watanabe (acnwata@ipc.shizuoka.ac.jp)
    Anwarul-Hassan Gilani (anwarhgilani@yahoo.com)

Version: 2 Date: 7 September 2014

Author's response to reviews: see over
To,

James Ellis D. Prozenko
Journal Editorial Office
BioMed Central

Dear James,

We would like to submit our revised manuscript MS: 1457526976127609 – “Studies on antidiarrhoeal, antispasmodic and bronchodilator activities of *Ipomoea turpethum* Linn” for publication as a research article in *BMC Complementary and Alternative Medicine*.

We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the manuscript and agree with submission to *BMC Complementary and Alternative Medicine*. The authors have no conflicts of interest to declare.

Thank you for your acceptance of this manuscript.

With personal regards

Anwar-ul-Hassan Gilani, PhD
Professor and Tandem Dean
College of Health Sciences
Mekelle University, PO Box 1871
Mekelle, Ethiopia
email: anwarhgilani@yahoo.com
Tel. mob: +251-932090467
web: www.mu.edu.et/CHS

Home Affiliation:
Aga Khan University Medical College,
Karachi, Pakistan www.aku.edu